Author’s response to reviews

Title: Self-medication among pregnant women attending antenatal clinic at Makongoro health centre in Mwanza, Tanzania: A challenge to health systems.

Authors:

Karol Marwa (carol_maro@yahoo.com)
Agnes Njalika (njalikaa@yahoo.com)
Deodatus Ruganuza (deorugz@gmail.com)
Deogratias Katabalo (dkatabalo@gmail.com)
Erasmus Kamugisha (erasmuskamugisha@yahoo.com)

Version: 5 Date: 29 Sep 2017

Author’s response to reviews:

Response to editorial board comments

Title: Self-medication among pregnant women attending antenatal clinic at Makongoro health centre in Mwanza, Tanzania: A challenge to health systems

Department of pharmacology, Catholic university of Health and Allied Sciences, Mwanza, Tanzania

Email: carol_maro@yahoo.com

25th September 2017

Editor, BMC Pregnancy and Child Birth

Dear editors,

I would like to thank you for constructive comments which have made the manuscript better than before.

Please find the letter showing authors response to the editorial board report as shown below using a YELLOW COLOUR.

It is my hope the manuscript will be published soon. Please contact me if any clarification is needed.
Sincerely yours,

Karol J. Marwa

Corresponding author.

<table>
<thead>
<tr>
<th>SNo.</th>
<th>Review’s comment</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>1.</td>
<td>Please amend your Limitation section with information about the lacking sensitivity analysis on women with full overview of the pregnancy; it should be also stated how/in what direction this factor may have affected the results.</td>
<td>The amendments have been done as suggested by the editor.</td>
</tr>
</tbody>
</table>